

# Item-based analysis of the effects of duloxetine in depression: a patient-level post hoc study

Downloaded from: https://research.chalmers.se, 2021-08-31 12:54 UTC

Citation for the original published paper (version of record):

Lisinski, A., Hieronymus, F., Näslund, J. et al (2020) Item-based analysis of the effects of duloxetine in depression: a patient-level post hoc study Neuropsychopharmacology, 45(3): 553-560 http://dx.doi.org/10.1038/s41386-019-0523-4

N.B. When citing this work, cite the original published paper.

research.chalmers.se offers the possibility of retrieving research publications produced at Chalmers University of Technology. It covers all kind of research output: articles, dissertations, conference papers, reports etc. since 2004. research.chalmers.se is administrated and maintained by Chalmers Library



## **ARTICLE** OPEN Item-based analysis of the effects of duloxetine in depression: a patient-level post hoc study

Alexander Lisinski<sup>1</sup>, Fredrik Hieronymus <sup>b</sup>, Jakob Näslund<sup>1</sup>, Staffan Nilsson<sup>2,3</sup> and Elias Eriksson<sup>1</sup>

Oft-cited trial-level meta-analyses casting doubt on the usefulness of antidepressants have been based on re-analyses of to what extent the active drug has outperformed placebo in reducing the sum score of the Hamilton Depression Rating Scale (HDRS-17sum) in clinical trials. Recent studies, however, suggest patient-level analyses of individual HDRS items to be more informative when assessing the efficacy of an antidepressant. To shed further light on both symptom-reducing and symptom-aggravating effects of a serotonin and noradrenaline reuptake inhibitor, duloxetine, when used for major depression in adults, we hence applied this approach to re-analyse data from 13 placebo-controlled trials. In addition, using patient-level data from 28 placebo-controlled trials of selective serotonin reuptake inhibitors (SSRIs), the response profile of duloxetine was compared to that of these drugs. Duloxetine induced a robust reduction in depressed mood that was not dependent on baseline severity and not caused by sideeffects breaking the blind. A beneficial effect on depressed mood was at hand already after one week; when outcome was assessed using HDRS-17-sum as effect parameter, this early response was however masked by a concomitant deterioration with respect to adverse event-related items. No support for a suicide-provoking effect of duloxetine was obtained. The response profile of duloxetine was strikingly similar to that of the SSRIs. We conclude that the use of HDRS-17-sum as effect parameter underestimates the true efficacy and masks an early effect of duloxetine on core symptoms of depression. No support for major differences between duloxetine and SSRIs in clinical profile were obtained.

Neuropsychopharmacology (2020) 45:553-560; https://doi.org/10.1038/s41386-019-0523-4

#### INTRODUCTION

The ongoing debate regarding the alleged ineffectiveness of antidepressants has to a large extent been based on trial-based meta-analyses using the sum score of the 17-item Hamilton Depression Rating Scale (HDRS-17-sum) as effect parameter [1]. The HDRS-17 has, however, been criticised for shortcomings likely to reduce the apparent difference between active drug and placebo: it is multidimensional, it includes items that are often absent at baseline, it includes non-specific symptoms often present also in non-depressed patients, and it may be contaminated by common antidepressant side effects [2-6]. Thus, in patient-level analyses comparing selective serotonin reuptake inhibitors (SSRIs) to placebo with respect to reduction in the first item of the HDRS-17, depressed mood, a considerably more consistent and robust superiority of active drug over placebo was revealed as compared to when HDRS-17-sum was used as effect parameter [5, 6].

One purpose of the present work was to use patient-level data to explore if these results can be extended to an antidepressant with a somewhat different pharmacological profile, the serotonin and noradrenalin reuptake inhibitor (SNRI) duloxetine [7]. Prompted by claims that a possible effect of antidepressants be restricted to the most severe cases [1, 8], or be merely the consequence of side effect breaking the blind in patients [9], these possibilities were also addressed, as was the issue of to what extent duloxetine may cause initial anxiety- or suicide-provoking effects. Whereas some of these aspects have been examined in previous meta-and mega-analyses [10–12], this is, to our knowl-edge, the most comprehensive patient-level post hoc analysis of duloxetine to date. Likewise, it is the most extensive analysis yet undertaken regarding possible differences between duloxetine and the SSRIs with respect to response profile.

#### MATERIALS AND METHODS

Data acquisition and participants

Patient-level data for 15 drug company-sponsored, placebocontrolled clinical trials regarding the treatment of depression in adults with duloxetine and using the HDRS-17 for assessment of efficacy were obtained from the Clinical Study Data Request (CSDR) website; whereas 10 of these were named in the FDA Approval Packages for duloxetine, five were post-marketing trials. We confirmed that we had access to all relevant trials by searching the FDA Approval Packages, the CSDR website, Eli Lilly's Clinical Study Results portal and Clinicaltrials.gov. For one trial named in the FDA Approval Package, HMAG, data were not available. For two of the trials, HMAH and HMAI, we only had access to information regarding HDRS-17-sum and the items depressed mood and suicidality; moreover, since these trials only included sub-therapeutic doses (5–20 mg), they were included in a

<sup>1</sup>Department of Pharmacology, Institute of Neuroscience and Physiology, Sahlgrenska Academy, University of Gothenburg, Gothenburg, Sweden; <sup>2</sup>Department of Pathology and Genetics, Institute of Biomedicine, Sahlgrenska Academy, University of Gothenburg, Gothenburg, Sweden and <sup>3</sup>Institute of Mathematical Sciences, Chalmers University of Technology, Gothenburg, Sweden

Correspondence: Elias Eriksson (elias.eriksson@neuro.gu.se)

Received: 9 May 2019 Revised: 12 August 2019 Accepted: 3 September 2019 Published online: 14 September 2019

554

sensitivity analysis only. While all remaining trials included adult patients aged at least 18, two of them, HMBV and HMFA, included elderly patients only and one, HMCB, only patients scoring a minimum of two points on the Brief Pain Inventory scale. For the comparison of duloxetine and the SSRIs with respect to item response profile, we also used patient-level data from 28 drug company-sponsored placebo-controlled trials of citalopram, paroxetine and sertraline; for details regarding this database, see ref. [5]. Since some duloxetine studies were so designed that neither patients nor investigators knew at which visit the administration of active treatment started (double-blind variable lead-in),[13] 486 patients displayed HDRS-17 ratings below 15 points at the start of active treatment. These were all retrospectively excluded by the investigators in the original studies [14-20], and so also in our primary analyses. In total, our primary analyses comprised 3575 subjects.

#### Statistical analysis

Linear mixed models including change score for the relevant measure (HDRS-17-sum or individual items) as the dependent variable, time (week), treatment (duloxetine or placebo), trial and the treatment\*time interaction as fixed factors, and baseline rating for the corresponding outcome measure as covariate, were used to assess the effect of treatment (duloxetine) on all included outcome measures. For mean scores of the suicidality item, the analysis comprising all subjects was complemented with analyses undertaken after splitting the cohort into those 18-24 and those ≥25 years of age, respectively. The model described above was expanded when assessing the possible interaction between HDRS-17-rated baseline severity and treatment and when comparing the relative efficacy of duloxetine to that of the SSRIs (see below). ANCOVA models on the observed cases population were used when addressing the association between side effects and treatment efficacy (see below).

For all linear mixed models, an unstructured covariance matrix was used to model within-patient errors; if the model did not converge, an autoregressive heterogeneous structure was used instead. Denominator degrees of freedom were estimated using the Kenward-Roger approximation. Effect sizes were calculated by dividing the least-squares mean differences for the relevant contrast by the root of the variance for the corresponding time point.

The possible relationship between baseline severity measured by HDRS-17-sum and outcome was assessed using a model similar to the one described above, but always including the baseline score for HDRS-17-sum (rather than the baseline score of the dependent variable) as well as the corresponding two-way (baseline score\*treatment, baseline score\*time) and three-way (baseline score\*treatment\*time) interactions. HDRS-17 sum, depressed mood, the sum of six core symptoms of depression (HDRS-6) [21] and the sum of the remaining 11 symptoms (non-HDRS-6) served as effect parameters [22]. The outcome of interest was the parameter estimates for the two-way interaction between HDRS-17-sum rated baseline severity and treatment since this parameter corresponds to the interaction effect at the endpoint visit (week 8), which was the reference category. In addition, patients at the extreme ends of the severity range, scoring  $\leq 18$ and  $\geq$ 27 respectively, i.e. the lowest proposed cut-off for moderate depression and the highest proposed for severe depression [23], were compared with respect to baseline symptomatology and with respect to endpoint effect sizes for HDRS-6, non-HDRS-6 and all individual items.

Since a possible association between side effects and outcome may be masked by patients with adverse events more often discontinuing early, and by potential side effects of antidepressants contaminating the HDRS-17-sum measure, the analysis addressing this issue comprised completers only and used depressed mood as effect parameter; a sensitivity analysis replacing depressed mood with HDRS-17-sum is, however, also presented. Presence of adverse events either during the first 2 weeks of treatment, or at any time until endpoint, was coded as a dichotomous variable (yes/no) [24]. Adverse events that had been reported already before the initiation of treatment were excluded. The three groups thus compared, using analysis of covariance (ANCOVA) with baseline values on the corresponding outcome measure as covariate and trial as fixed factor, were placebo-treated patients, duloxetine-treated patients with side effects and duloxetine-treated patients without side effects. For assessing the possible association between adverse event severity and reduction in depressed mood, only duloxetine-treated patients with adverse events were included, with the highest side effect severity, coded as mild, moderate or severe, being included in an ANCOVA as described above.

Finally, the item response profile of duloxetine was compared to that of the SSRIs. To this end, patient-level data from the duloxetine trials on the one hand, and patient-level data from 28 relevant drug company-sponsored trials conducted for citalopram, paroxetine and sertraline [5] on the other, were used to assess the relative effect size for each HDRS item using the sum of all other HDRS items as covariate in two separate linear mixed models. For these analyses week 6 served as endpoint since many of the SSRI studies had been of only six weeks duration. In those of the SSRI trials in which fluoxetine had been used as an active comparator, this treatment arm was also included in the analyses. Student's *t*-test was used to compare subjects treated with duloxetine and an SSRI, respectively, with regard to the relative mean active drug versus placebo differences.

For all analyses, significance tests were two-tailed with an alpha level of .05. No correction for multiple testing was undertaken.

Remote desktop access to the Clinical Trial Data Transparency environment was provided by CSDR through SAS Solutions OnDemand, using SAS version 9.4 (SAS Institute, Cary, NC, USA).

#### Sensitivity analyses

To assess if the decision to exclude two low-dose trials impacted the outcome, sensitivity analyses comprising all doses were undertaken. Similarly, to explore the possible influence of excluding patients scoring <15 points on the HDRS-17 at baseline, sensitivity analyses including these patients were performed. A final sensitivity analysis addressed the possible impact of side effects on outcome using the HDRS-17-sum rather than depressed mood as effect parameter. For all sensitivity analyses, the same statistical models as in the corresponding main analysis were used.

#### Ethics

The Regional Ethics Review Board of Gothenburg, Sweden, issued an advisory opinion stating no objection to the conduct of this study.

#### RESULTS

#### Effect sizes for individual items

Baseline characteristics for all included trials are displayed in Supplementary table 1. Mean values of HDRS-17-sum and depressed mood at baseline were 21.3 points (duloxetine: 21.3 points, placebo: 21.3 points) and 2.71 points (duloxetine: 2.71 points, placebo: 2.69 points), respectively. Effect sizes and levels of significance for duloxetine with respect to HDRS-17-sum as well as all individual items for weeks 1, 6 and 8 are presented in Table 1. Three studies however lacked an evaluation at week 6 and 8; for these observations, week 7 (HMCB) or week 9 (HMBHa + b) were used as endpoint observations, and week 5 (HMCB) or 7 (HMBHa + b) as replacement for week 6 data. Week-for-week data for HDRS-17-sum, depressed mood and psychic anxiety are presented in Fig. 1a–c and for suicidality in Fig. 2a, b.

Table 1. Baseline means, effect sizes and p-values for HDRS-17-sum and individual items							
Measure of efficacy (scoring range)	Baseline mean	s.d.	ES Week 1	<i>p</i> -value	ES Week 6	<i>p</i> -value	ES Week 8
HDRS-17-sum	21.3	3.9	0.05	0.16	0.37	<0.001	0.37
Depressed mood (0–4)	2.7	0.6	0.20	<0.001	0.46	<0.001	0.44
Feelings of guilt (0–4)	1.6	0.8	0.10	0.004	0.27	<0.001	0.28
Suicidal ideation (0–4)	0.6	0.8	0.15	<0.001	0.22	<0.001	0.23
Insomnia, early (0–2)	1.2	0.8	0.05	0.14	0.06	0.18	0.09
Insomnia, middle (0–2)	1.3	0.8	-0.08	0.02	0.03	0.45	0.03
Insomnia, late (0–2)	1.1	0.8	-0.02	0.60	0.18	<0.001	0.13
Work & activities (0–4)	2.7	0.6	0.05	0.13	0.30	<0.001	0.32

0.8

09

0.8

0.8

0.6

0.6

0.8

0.9

0.5

0.4

0.04

0.08

0.23

-0.04

-0.30

-0.02

-0.10

0.04

-0.22

-0.05

0.29

0.02

<0.001

0.25

< 0.001

0.66

0.004

0.27

<0.001

0.15

0.29

0.11

0.42

0.09

0.08

0.25

0.05

0.22

-0.03

0.04

< 0.001

0.01

< 0.001

0.04

0.09

< 0.001

0.29

< 0.001

0.55

0.34

1.0

0.9

2.0

1.5

0.6

1.6

1.2

1.0

0.2

0.1

(b) (a) (c) 0.00 0.0 Placebo (n=1345) Dulovetine (r =2230 SEM (HDRS-17-sum) -3.75 -0.5 -0.5 ESEM (HDRS ps HDRS SEM -1.0 -1.0 -7.50 -1.5 -2.0 -15.00 -2.0 Ó 2 4 à Weel

Fig. 1 a-c Week-by-week mean change for HDRS-17-sum (a), the depressed mood item (b) and the psychic anxiety item (c) for duloxetineand placebo-treated subjects, respectively. Lines represent estimated means from a linear mixed model. Effect sizes at week 1, 2, 4, 6 and 8, respectively: **a** 0.05, 0.20, 0.27, 0.37, 0.37, **b** 0.20, 0.31, 0.37, 0.46, 0.44 **c**; 0.23, 0.29, 0.34, 0.42, 0.43. *n.s.* non-significant (*p* = 0.16); \*\*\* = *p* < 0.001

The effect size for the comparison of duloxetine with placebo was higher for depressed mood and psychic anxiety than for HDRS-17-sum at all time points (Fig. 1a-c). A significant superiority of duloxetine over placebo was found for the items depressed mood, guilt, suicidality, agitation and psychic anxiety, but not for HDRS-17-sum, at week one. Ratings of the items middle insomnia, gastrointestinal symptoms, sexual dysfunction and loss of weight were higher in duloxetine-treated patients after one week of treatment but not at endpoint (Table 1). Analyses of the rating of suicidality after splitting the population into those 18-24 or  $\geq 25$ years of age, respectively, yielded a significant superiority of duloxetine from week one and onwards in those  $\geq$  25 years of age (n = 3 318). In those 18–24 years of age (n = 257), patients on duloxetine displayed numerically but non-significantly lower ratings (Fig. 2a, b).

Relation between initial severity and outcome

The interaction between baseline severity based on HDRS-17-sum and treatment was non-significant when using the HDRS-17-sum (beta 0.10, standard error of the mean (SEM) 0.07, p = 0.11), depressed mood (beta 0.01, SEM 0.01, p = 0.56) or the sum of 6 core symptoms of depression (beta 0.02, SEM 0.04, p = 0.50) as outcome measure, but significant in the direction of larger drugplacebo differences with increasing severity when analysing the

Retardation (0-4)

Anxiety, psychic (0-4)

Anxiety, somatic (0-4)

Genital symptoms (0-2)

Hypochondriasis (0-4)

Loss of weight (0-2)

Insight (0-2)

Somatic symptoms, gastrointestinal (0-2)

Somatic symptoms, general (0-2)

s.d. standard deviation, ES effect size

Agitation (0-4)

p-value

< 0.001

< 0.001

< 0.001

< 0.001

0.02

0.52

0.001

< 0.001

< 0.001

0.01

< 0.001

0.001

0.03

< 0.001

0.01

< 0.001

0.19

0.07

0.22

0 10

0.43

0.13

0.09

0.19

0.10

0.22

-0.05

0.07

Item-based analysis of the effects of duloxetine in depression: a... A Lisinski et al.



**Fig. 2 a**, **b** Week-by-week mean change for the suicidality item for duloxetine and placebo-treated subjects, respectively, in subjects aged 25 years and above (**a**) subjects aged 18–24 years (**b**). Lines represent estimated means from a linear mixed model. Baseline scores **a** 0.62 points, standard deviation (SD) 0.76; **b** 0.53 points, SD 0.70. Effect sizes at week 1, 2, 4, 6 and 8, respectively: **a** 0.15, 0.23, 0.26, 0.23, 0.24; **b** 0.22, 0.22, 0.13, 0.25, 0.20. *n.s.* non-significant (p = 0.10, 0.11, 0.42, 0.16, 0.21); \*\*\*p < 0.001

sum of the remaining 11 symptoms (beta 0.08, SEM 0.03, p = 0.02). Baseline symptomatology and endpoint effect sizes for the extreme severity groups ( $\leq 18$  and  $\geq 27$ , respectively) are presented in Supplementary Table 2 and Supplementary Fig. 1a, b. While the mean non-HDRS-6 sum score was 139% higher in the high severity group, the HDRS-6 sum score was merely 63% higher. Likewise, while the effect sizes for the HDRS-6 sum score and for individual items comprising this subscale were similar in the two extreme severity groups, the effect sizes for the sum score of the non-HDRS-6 items, and for several of the items within this subscale, were notably lower in the non-severe group.

#### Relation between side effects and outcome

Both duloxetine-treated patients not reporting early side effects and those not reporting any side effects throughout the trial displayed a significant reduction in depressed mood when compared to those treated with placebo (Fig. 3a, b). This response was moderately but significantly larger in those with side effects, but there was no impact of side effect severity on outcome (Supplementary Fig. 2).

### Symptom response patterns in patients treated with duloxetine or SSRIs

A comparison of all duloxetine-treated subjects with SSRI-treated patients within the development programs for citalopram, paroxetine and sertraline with respect to the relative reduction of individual items after adjusting for overall response revealed strikingly similar profiles, with no significant differences between the groups (Fig. 4).

#### Sensitivity analyses

The outcome of the sensivity analyses for item-wise comparisons of duloxetine versus placebo including also low-dose studies or including all patients regardless of initial HDRS score were similar to that of the main analyses (Supplementary tables 3 and 4, respectively). The sensitivity analysis regarding the possible relationship between side effects and outcome when using HDRS-17-sum rather than depressed mood as outcome measure yielded no influence of side effects on the response to treatment (Supplementary Fig. 3a, b).

#### DISCUSSION

Based on trial-level meta-analyses using HDRS-17-sum as effect parameter, oft-cited authors have claimed that the effect of antidepressants is minute and clinically relevant, if at all, in cases of severe depression only [1, 8]. However, invoking results from patient-level, item-based post hoc analyses of placebo-controlled trials, we have previously suggested the apparently poor effect of the SSRIs to be an artefact caused by the use of a partly misleading effect parameter; thus, when exploring the effect of SSRIs on the depressed mood item, rather than on HDRS-17-sum, a consistent and robust superiority of SSRIs over placebo was revealed [5]. We now report the results of similar, patient-level, item-based post hoc analyses of a non-SSRI antidepressant, duloxetine, confirming this observation. Thus, in spite of other methodological problems associated with antidepressant trials, that should be expected to lead to an underrating of the efficacy of the active drug [5, 25], the effect size for the reduction of depressed mood obtained in the present analysis, 0.44, is in the medium range and well on par with that displayed by drugs used for somatic conditions [26] (and higher than that obtained for HDRS-17-sum). Moreover, the beneficial effect of duloxetine was not restricted to those with severe depression and not dependent on side effects breaking the blind. A comparison of patients treated with duloxetine and SSRIs, respectively, revealed a strikingly similar item response pattern. The outcome of sensitivity analyses including also low-dose studies or patients with low baseline scores was essentially the same as that of the main analyses.

The previous claim that antidepressants be ineffective in mild and moderate cases [1] has prompted authorities in many countries to recommend non-pharmacological treatment for non-severe depression. Our analysis, however, revealed the duloxetine-induced reduction in depressed mood and other core symptoms of depression in patients at the lower end of the severity spectrum to be largely on par with that observed in the most severe cases; in line with this, no significant interaction between baseline severity and treatment was observed when addressing outcome using these measures. For the reduction in HDRS-17-sum, however, a non-significant tendency for an influence of severity on response was observed, which can largely

Item-based analysis of the effects of duloxetine in depression: a... A Lisinski et al.



**Fig. 3 a**, **b** Estimated endpoint means and effect sizes for the depressed mood item in placebo-treated subjects and in duloxetine-treated subjects with or without adverse events **a** only counting initial adverse events (reported during week 1–2 and **b** counting any adverse event throughout the trial. Placebo: n = 998; duloxetine without early adverse events: n = 713; duloxetine with early adverse events: n = 1 029; duloxetine without any adverse event throughout the trial: n = 449; duloxetine with any adverse event throughout the trial: n = 1 293. Effect sizes: **a** duloxetine with adverse events: 0.46; duloxetine without adverse events vs placebo: 0.34, duloxetine with adverse events vs placebo: 0.36; duloxetine with adverse events vs placebo: 0.12. \*p = 0.02 (**a**), p = 0.04 (**b**); \*\*\*p < 0.001

be explained by lower effect sizes for non-core symptoms in the non-severe group; thus, the effect size for the sum of the non-core items was indeed significantly associated with baseline severity. When interpreting this observation, it should be noted that the rating of non-core symptoms of depression, considerably more so than that of core symptoms, often was low or zero at baseline in cases defined as non-severe, hence leaving little room for improvement, but ample room for aggravation also in patients in remission should the item capture possible side effects of the given treatment. We hence suggest that the widely disseminated claim that antidepressants are useless for non-severe depression is misleading and the result of the unfortunate use of an outcome measure comprising many items that are often absent at baseline in the non-severe group. Though not comprising any interaction analyses corresponding to those here reported, a previous report from Shelton and co-workers [12], based on post hoc analyses of four duloxetine trials, was largely in line with our findings.

Another widely propagated claim aimed to cast doubt on the usefulness of antidepressants is that the superiority of active drug over placebo observed in antidepressant trials be merely a consequence of enhanced expectation of improvement due to side-effects breaking the blind [9]. The finding that the effect size for the reduction in depressed mood (unlike that for the reduction in HDRS-17-sum) was somewhat larger in duloxetine-treated patients reporting side effects than in those not reporting side effects could be regarded as support for this assumption; needless to say, this association may, however, also be the consequence of inter-individual differences in dose and/or duloxetine metabolism that, by influencing serum levels, may impact both response and side effects in the same direction. More importantly, the theory that side effects are the major factor for active treatment outperforming placebo was rebutted by the

Neuropsychopharmacology (2020) 45:553 – 560

observation of a marked superiority of duloxetine over placebo also in patients reporting no adverse events. Likewise, no support was obtained for the suggestion by Kirsch that adverse event severity be positively associated with response [9]. Both the latter observations are in line with what we previously reported for two SSRIs [24].

Attempts to reveal the mechanism of action of SSRIs and SNRIs, beyond the inhibition of transmitter reuptake, have often been based on the postulate that no clinical effect is observed until the treatment has been ongoing for a number of weeks [27]. However, the failure to detect an early effect of drug treatment using the HDRS-17-sum may be due to early side-effects of treatment contaminating the rating, hence masking an early improvement with respect to core symptoms of depression such as depressed mood [5, 6]. In this vein, while there was no difference between treatment groups with respect to HDRS-17-sum at week 1, duloxetine-treated patients reported less severity with respect to several cardinal symptoms of depression already at this time point, including depressed mood, but significantly more middle insomnia, gastrointestinal symptoms, sexual dysfunction and loss of weight. We hence gained support for our previous conclusion, based on similar post hoc analyses of SSRI trials [6], that the antidepressant action of amine reuptake inhibitors, though slow to reach its maximum, does commence shortly after the onset of treatment. Our results are also in line with previous reports suggesting duloxetine to outperform placebo with respect to individual items already at week 1 [12, 27], as well as with a report on the use of venlafaxine in generalised anxiety disorder indicating early side effects to mask an early improvement with respect to certain symptoms (including anxiety) [28].

Antidepressants displaying high affinity for the serotonin transporter, such as clomipramine and the SSRIs, may cause an

557



**Fig. 4** Relative effect sizes versus placebo for duloxetine and four SSRIs (grouped) using the sum of all other HDRS-17 items as covariate. Placebo (duloxetine studies): n = 1 345; duloxetine: n = 2 230; placebo (SSRI studies): n = 2 581; SSRIs: n = 5 681 (citalopram n = 744; fluoxetine n = 754; paroxetine n = 2 981; sertraline n = 1 202). HDRS-17 items and p-values for the contrast duloxetine vs SSRIs: 1 = depressed mood (p = 0.63), 2 = feelings of guilt (p = 0.35), 3 = suicidality (p = 0.89), 4 = early insomnia (p = 0.59), 5 = middle insomnia (p = 0.59), 6 = late insomnia (p = 0.63), 1 = somatic anxiety (p = 0.82), 8 = psychomotor retardation (p = 0.49), 9 = psychomotor agitation (p = 0.26), 10 = psychic anxiety (p = 0.65), 11 = somatic anxiety (p = 0.93), 12 = gastrointestinal symptoms (p = 0.23), 13 = general somatic symptoms (p = 0.39), 14 = sexual dysfunction (p = 0.54), 15 = hypochondriasis (p = 0.45), 16 = weight change (p = 0.80) and 17 = insight (p = 0.78)

initial increase in anxiety in susceptible patients [29, 30]. Our previous attempt to clarify the early net effect of SSRIs on HDRS-17-assessed anxiety revealed psychic anxiety and agitation to be moderately lower, but somatic anxiety to be moderately higher, after the first week of treatment in those administered an SSRI [31]. The corresponding outcome for duloxetine was well in line with that observed for the SSRIs: a significant reduction in psychic anxiety and agitation but not in somatic anxiety at week 1 [11]. However, while the rating of somatic anxiety at week 1 was numerically higher for duloxetine-treated patients, this difference was not significant, as It was for the SSRIs. At week 6 and at endpoint, all three items were in favour of duloxetine. It should, however, be noted that a complete picture of a possible anxietyprovoking effect of an antidepressant requires an analysis also of the reporting of anxiety-related adverse events [31], particularly in patients discontinuing before the week 1 visit, and that no such assessment was included in this study.

Reports suggesting SSRIs to be suicide-provoking, and particularly so at the start of treatment [32] and in young subjects [33], have previously prompted us [34] and others [35] to address this issue by analysing HDRS-17-assessed suicidality at the week one visit following initiation of treatment after splitting the population into young adults and adults, respectively. While these previous studies have shown SSRIs [34, 35], as well as an SNRI, venlafaxine [35], to cause a net reduction in suicidality from week 1 or 2 and onwards in subjects  $\geq$ 25 years of age, they have failed to demonstrate a corresponding effect in subjects below the age of 25, but also without providing any evidence of a suicidalityenhancing effect in this age group. A previous comprehensive post hoc analysis regarding the possible impact of duloxetine on self-rated suicidal ideation suggested the net effect of the drug on the suicidality item of the HDRS to be beneficial, but did not

specifically address the issue of possible early effects or age [10]. The present results were similar to those previously obtained for SSRIs and venlafaxine; the mean rating of suicidality in duloxetinetreated subjects compared to those on placebo was hence significantly lower in subjects ≥25 years already from week one and onwards. In the younger age group, a corresponding superiority of duloxetine was also at hand from week one and onwards, but never reached the level of statistical significance; when interpreting this difference between the two age groups, it should, however, be noted that the number of subjects aged 18–24 was much lower (n = 257) than that of the older subjects (n= 3318). While these data do not exclude the possibility that duloxetine may exert a paradoxical suicide-provoking effect in susceptible individuals, they lend no support for the assumption that the net effect of the drug on suicidality, at the first week of treatment or at endpoint, be harmful, either in those below or above the age of 25. Since no children or adolescents were included, our data, however, permit no conclusions with respect to the possible effect of duloxetine on suicidality in these age groups.

Duloxetine displaying affinity not only for the transporter of serotonin, but also for that of noradrenaline [7], has prompted previous writers to suggest that this drug displays a somewhat different response profile as compared to that of the SSRIs. In line with this, a previous post hoc analysis from Mallinckrodt and co-workers, based on six trials comprising a duloxetine arm, an SSRI arm and placebo, suggested duloxetine to outperform the SSRIs with respect to a number of items (work and activities, psychomotor retardation, genital symptoms and hypochondriasis) [36]. To some extent, the interpretation of the outcome of these head-to-head comparisons is, however, marred by the fact that the SSRI dose applied in three of them, as acknowledged by the

authors, may have been suboptimal [37, 38]. We hence used a different path to explore the same issue, i.e. to compare the relative item response profile (after adjustment for overall response) of the duloxetine-treated patients included in the dataset on which this paper is based with that of the SSRI-treated patients included in the development programs for citalopram, paroxetine and sertraline (n = 8262) [5]. This analysis revealed a strikingly similar response profile for the SSRIs and duloxetine, respectively, hence arguing against any major differences between these two treatments. While it should be underlined that adequately powered head-to-head trials using equipotent doses are required to definitely settle the issue of possible differences in efficacy or effect profile between two treatments. this lack of apparent differences in response profile is somewhat surprising; since duloxetine but not the SSRIs is reported to inhibit the noradrenaline transporter (NET), certain dissimilarities in clinical response profile might indeed have been expected. One possible explanation to this similarity of SSRIs and an SNRI, respectively, could be that the influence of duloxetine on NET in the human brain at the dosage used to treat depression may be lower than usually assumed; recent positron emission tomography studies thus indicate that duloxetine at clinically relevant dosage blocks less than 40% of NET but around 80% of the serotonin transporter (SERT) [39]. It is also possible that the contribution of additional NET inhibition to an antidepressant response obtained by SERT inhibitors has been overrated, and/or that the response pattern observed reflects the nature of the disorder rather than of the given treatment, i.e. that a similar response profile should be expected for any effective antidepressant regimen when compared to placebo. Similar studies comparing the response profile of SSRIs and duloxetine with that of antidepressants not influencing SERT or NET might shed further light on this issue. Finally, when interpreting the present data, it should be considered that there may indeed exist differences between duloxetine and SSRIs that are not captured by any of the HDRS-17 items; duloxetine has, for example, been attributed superior efficacy for the treatment of pain [40], which is an effect that is not necessarily reflected by the HDRS-17.

The present analysis differs from the majority (though not all) of previous meta-analyses in the antidepressant field by being conducted on patient- rather than trial-level, and by analysing individual symptoms rather than the sum score of several items. We believe that such an approach, which is in line with the RDoC focus on symptoms and dimensions rather than diagnoses [41], to be particularly useful when analysing a condition such as depression, where patients differ considerably with respect to symptom profile at baseline. Of note is that a number of widely spread assumptions regarding the antidepressant drugs, such as their symptom-reducing effect being secondary to side effects [9], or the antidepressant effect being restricted to those with severe depression [1, 8], were based on trial-level, HDRS-17-sum-based meta-analyses, which have, however, now been rebutted by analyses conducted on the patient level [22] and/or being itembased [24, 42] (including the present one).

To conclude, item-based analyses of duloxetine trials revealed this drug to exert a robust reduction in depressed mood and other core symptoms of depression that, for several of these, was significant already after 1 week of treatment. The effect of duloxetine was not restricted to those with high HDRS-17-sum at baseline and not secondary to side effects; moreover, no support for a suicide-provoking effect was obtained. Comparisons of the effect profiles of duloxetine and SSRIs, respectively, revealed a strikingly similar picture.

#### FUNDING AND DISCLOSURE

The study was supported by the Swedish Medical Research Council, AFA Insurance, the Swedish Brain Foundation and by a Item-based analysis of the effects of duloxetine in depression: a... A Lisinski et al.

559

grant from the Swedish state under the agreement between the Swedish government and the county councils (ALF). FH has received speaker's fees from Servier and H Lundbeck. EE has been on advisory boards and/or received speaker's honoraria and/or research grants from Eli Lilly, Servier, GSK, H Lundbeck and Janssen Cilag. The remaining authors declare no competing interests.

#### ACKNOWLEDGEMENTS

Patient-level data for duloxetine were obtained from the Clinical Study Data Request (CSDR) website. Patient-level data for citalopram, paroxetine and sertraline were obtained from Lundbeck, GSK and Pfizer, respectively.

#### **ADDITIONAL INFORMATION**

Supplementary Information accompanies this paper at (https://doi.org/10.1038/ s41386-019-0523-4).

**Publisher's note** Springer Nature remains neutral with regard to jurisdictional claims in published maps and institutional affiliations.

#### REFERENCES

- Kirsch I, Deacon BJ, Huedo-Medina TB, Scoboria A, Moore TJ, Johnson BT. Initial severity and antidepressant benefits: a meta-analysis of data submitted to the Food and Drug Administration. PLoS Med. 2008;5:e45.
- Bagby RM, Ryder AG, Schuller DR, Marshall MB. The Hamilton Depression Rating Scale: has the gold standard become a lead weight? Am J Psychiatry. 2004;161:2163–77.
- Bech P, Allerup P, Gram LF, Reisby N, Rosenberg R, Jacobsen O, et al. The Hamilton depression scale. Evaluation of objectivity using logistic models. Acta Psychiatr Scand. 1981;63:290–9.
- Fried El, van Borkulo CD, Epskamp S, Schoevers RA, Tuerlinckx F, Borsboom D. Measuring depression over time... Or not? Lack of unidimensionality and longitudinal measurement invariance in four common rating scales of depression. Psychol Assess. 2016;28:1354–67.
- Hieronymus F, Emilsson JF, Nilsson S, Eriksson E. Consistent superiority of selective serotonin reuptake inhibitors over placebo in reducing depressed mood in patients with major depression. Mol Psychiatry. 2016;21:523–30.
- Hieronymus F, Nilsson S, Eriksson E. A mega-analysis of fixed-dose trials reveals dose-dependency and a rapid onset of action for the antidepressant effect of three selective serotonin reuptake inhibitors. Transl Psychiatry. 2016;6:e834.
- Wong DT. Duloxetine (LY 248686): an inhibitor of serotonin and noradrenaline uptake and an antidepressant drug candidate. Expert Opin Investig Drugs. 1998;7:1691–9.
- Khan A, Leventhal RM, Khan SR, Brown WA. Severity of depression and response to antidepressants and placebo: an analysis of the Food and Drug Administration database. J Clin Psychopharmacol. 2002;22:40–5.
- 9. Kirsch I. The emperor's new drugs: exploring the antidepressant myth. New York: Basic Books; 2010.
- Acharya N, Rosen AS, Polzer JP, D'Souza DN, Perahia DG, Cavazzoni PA, et al. Duloxetine: meta-analyses of suicidal behaviors and ideation in clinical trials for major depressive disorder. J Clin Psychopharmacol. 2006;26:587–94.
- Dunner DL, Goldstein DJ, Mallinckrodt C, Lu Y, Detke MJ. Duloxetine in treatment of anxiety symptoms associated with depression. Depress Anxiety. 2003;18:53–61.
- Shelton RC, Prakash A, Mallinckrodt CH, Wohlreich MM, Raskin J, Robinson MJ, et al. Patterns of depressive symptom response in duloxetine-treated outpatients with mild, moderate or more severe depression. Int J Clin Pract. 2007;61:1337–48.
- Faries DE, Heiligenstein JH, Tollefson GD, Potter WZ. The double-blind variable placebo lead-in period: results from two antidepressant clinical trials. J Clin Psychopharmacol. 2001;21:561–8.
- Detke MJ, Wiltse CG, Mallinckrodt CH, McNamara RK, Demitrack MA, Bitter I. Duloxetine in the acute and long-term treatment of major depressive disorder: a placebo- and paroxetine-controlled trial. Eur Neuropsychopharmacol. 2004;14:457–70.
- Perahia DG, Wang F, Mallinckrodt CH, Walker DJ, Detke MJ. Duloxetine in the treatment of major depressive disorder: a placebo- and paroxetine-controlled trial. Eur Psychiatry. 2006;21:367–78.
- Goldstein DJ, Lu Y, Detke MJ, Wiltse C, Mallinckrodt C, Demitrack MA. Duloxetine in the treatment of depression: a double-blind placebo-controlled comparison with paroxetine. J Clin Psychopharmacol. 2004;24:389–99.

- 560
- Goldstein DJ, Mallinckrodt C, Lu Y, Demitrack MA. Duloxetine in the treatment of major depressive disorder: a double-blind clinical trial. J Clin Psychiatry. 2002;63:225–31.
- Detke MJ, Lu Y, Goldstein DJ, Hayes JR, Demitrack MA. Duloxetine, 60 mg once daily, for major depressive disorder: a randomized double-blind placebo-controlled trial. J Clin Psychiatry. 2002;63:308–15.
- Detke MJ, Lu Y, Goldstein DJ, McNamara RK, Demitrack MA. Duloxetine 60 mg once daily dosing versus placebo in the acute treatment of major depression. J Psychiatr Res. 2002;36:383–90.
- Brannan SK, Mallinckrodt CH, Brown EB, Wohlreich MM, Watkin JG, Schatzberg AF. Duloxetine 60 mg once-daily in the treatment of painful physical symptoms in patients with major depressive disorder. J Psychiatr Res. 2005;39:43–53.
- 21. Bech P, Gram LF, Dein E, Jacobsen O, Vitger J, Bolwig TG. Quantitative rating of depressive states. Acta Psychiatr Scand. 1975;51:161–70.
- Furukawa TA, Maruo K, Noma H, Tanaka S, Imai H, Shinohara K, et al. Initial severity of major depression and efficacy of new generation antidepressants: individual participant data meta-analysis. Acta Psychiatr Scand. 2018;137:450–58.
- Kriston L, von Wolff A. Not as golden as standards should be: interpretation of the Hamilton Rating Scale for Depression. J Affect Disord. 2011;128:175–7.
- Hieronymus F, Lisinski A, Nilsson S, Eriksson E. Efficacy of selective serotonin reuptake inhibitors in the absence of side effects: a mega-analysis of citalopram and paroxetine in adult depression. Mol Psychiatry. 2018;23:1731–6.
- Fountoulakis KN, McIntyre RS, Carvalho AF. From randomized controlled trials of antidepressant drugs to the meta-analytic synthesis of evidence: methodological aspects lead to discrepant findings. Curr Neuropharmacol. 2015;13:605–15.
- Leucht S, Hierl S, Kissling W, Dold M, Davis JM. Putting the efficacy of psychiatric and general medication into perspective: review of meta-analyses. Br J Psychiatry. 2012;200:97–106.
- 27. Hirschfeld RM, Mallinckrodt C, Lee TC, Detke MJ. Time course of depressionsymptom improvement during treatment with duloxetine. Depress Anxiety. 2005;21:170–7.
- Stahl SM, Ahmed S, Haudiquet V. Analysis of the rate of improvement of specific psychic and somatic symptoms of general anxiety disorder during long-term treatment with venlafaxine ER. CNS Spectr. 2007;12:703–11.
- Sinclair LI, Christmas DM, Hood SD, Potokar JP, Robertson A, Isaac A, et al. Antidepressant-induced jitteriness/anxiety syndrome: systematic review. Br J Psychiatry. 2009;194:483–90.
- Pohl R, Yeragani VK, Balon R, Lycaki H. The jitteriness syndrome in panic disorder patients treated with antidepressants. J Clin Psychiatry. 1988;49:100–4.
- Näslund J, Hieronymus F, Emilsson JF, Lisinski A, Nilsson S, Eriksson E. Incidence of early anxiety aggravation in trials of selective serotonin reuptake inhibitors in depression. Acta Psychiatr Scand. 2017;136:343–51.
- 32. Jick H, Kaye JA, Jick SS. Antidepressants and the risk of suicidal behaviors. JAMA. 2004;292:338-43.
- Barbui C, Esposito E, Cipriani A. Selective serotonin reuptake inhibitors and risk of suicide: a systematic review of observational studies. CMAJ. 2009;180:291–7.

- Näslund J, Hieronymus F, Lisinski A, Nilsson S, Eriksson E. Effects of selective serotonin reuptake inhibitors on rating-scale-assessed suicidality in adults with depression. Br J Psychiatry. 2018;212:148–54.
- 35. Gibbons RD, Brown CH, Hur K, Davis J, Mann JJ. Suicidal thoughts and behavior with antidepressant treatment: reanalysis of the randomized placebo-controlled studies of fluoxetine and venlafaxine. Arch Gen Psychiatry. 2012;69:580–7.
- Mallinckrodt CH, Prakash A, Houston JP, Swindle R, Detke MJ, Fava M. Differential antidepressant symptom efficacy: placebo-controlled comparisons of duloxetine and SSRIs (fluoxetine, paroxetine, escitalopram). Neuropsychobiology. 2007;56:73–85.
- Hayasaka Y, Purgato M, Magni LR, Ogawa Y, Takeshima N, Cipriani A, et al. Dose equivalents of antidepressants: Evidence-based recommendations from randomized controlled trials. J Affect Disord. 2015;180:179–84.
- Jakubovski E, Varigonda AL, Freemantle N, Taylor MJ, Bloch MH. Systematic review and meta-analysis: dose-response relationship of selective serotonin reuptake inhibitors in major depressive disorder. Am J Psychiatry. 2016; 173:174–83.
- Moriguchi S, Takano H, Kimura Y, Nagashima T, Takahata K, Kubota M, et al. Occupancy of norepinephrine transporter by duloxetine in human brains measured by positron emission tomography with (S,S)-[18F]FMeNER-D2. Int J Neuropsychopharmacol. 2017;20:957–62.
- 40. Lee YC, Chen PP. A review of SSRIs and SNRIs in neuropathic pain. Expert Opin Pharmacother.2010;11:2813–25.
- Morris SE, Cuthbert BN. Research Domain Criteria: cognitive systems, neural circuits, and dimensions of behavior. Dialogues Clin Neurosci. 2012; 14:29–37.
- Hieronymus F, Lisinski A, Nilsson S, Eriksson E. Influence of baseline severity on the effects of SSRIs in depression: an item-based, patient-level post-hoc analysis. Lancet Psychiatry. 2019;6:745–52.

**Open Access** This article is licensed under a Creative Commons Attribution 4.0 International License, which permits use, sharing, adaptation, distribution and reproduction in any medium or format, as long as you give appropriate credit to the original author(s) and the source, provide a link to the Creative Commons license, and indicate if changes were made. The images or other third party material in this article are included in the article's Creative Commons license, unless indicated otherwise in a credit line to the material. If material is not included in the article's Creative Commons license and your intended use is not permitted by statutory regulation or exceeds the permitted use, you will need to obtain permission directly from the copyright holder. To view a copy of this license, visit http://creativecommons. org/licenses/by/4.0/.

© The Author(s) 2019